Title: Web-based tools can be used reliably to detect patients with major depressive disorder and subsyndromal depressive symptoms

Authors:

Chao-Cheng Lin (linchri@gmail.com)
Ya-Mei Bai (ymbi@mail2000.com.tw)
Chia-Yih Liu (liucy752@adm.cgmh.com.tw)
Mei-Chun Hsiao (norase@adm.cgmh.org.tw)
Jen-Yeu Chen (sms0604@mail2000.com.tw)
Shih-Jen Tsai (sjtsai@vghtpe.gov.tw)
Wen-Chen Ouyang (d88904@mail.cnpc.gov.tw)
Chia-hsuan Wu (erinwu1969@yahoo.com)
Yu-Chuan Li (jack@tmu.edu.tw)

Version: 6 Date: 25 March 2007

Author's response to reviews: see over
Dear the Editor:

Thank you for the acceptance of our manuscript for publication in BMC Psychiatry. We have revised the format of our manuscript according to your suggestions:

1. Keywords and short title were removed from the title page.

2. Website addresses were removed from the text, and cited from the reference list.

3. The resolution of Figure 1 has been improved and the image was cropped as closely as possible to minimise white space around the image.

Additional 2 corrections have been made:

1. In abstract, “Mandarin Chinese” was changed to “traditional Chinese”, which is a more precise description of Chinese written language.

2. In discussion, a typo on the 3rd line of 2nd paragraph was corrected (“good” changed to “fair”).

Sincerely yours,

Chao-Cheng Lin, M.D.
Department of Psychiatry,
National Taiwan University Hospital
Email: linchri@gmail.com